Author’s response to reviews

Title: Maternal plasma levels of oxytocin during physiological childbirth – a systematic review with implications for uterine contractions and central actions of oxytocin

Authors:

Kerstin Uvnäs Moberg (k_uvnas_moberg@hotmail.com)
Marie Berg (marie.berg@fhs.gu.se)
Anette Ekström (anette.ekstrom@his.se)
Sarah Buckley (dr.sarahjbuckley@gmail.com)
Soo Downe (SDowne@uclan.ac.uk)
Eleni Hadjigeorgiou (eleni.hadjigeorgiou@cut.ac.cy)
Bogumila Kielbratowska (bogumila.kielbratowska@gumed.edu.pl)
Alicja Kotlowska (kotlowska@gumed.edu.pl)
Luise Lengler (luise.lengler@t-online.de)
Fatima Leon-Larios (fatimaleon@us.es)
Bengt Lindström (bengtblind@hotmail.com)
Claudia Meier Magistretti (claudia.meiermagistretti@hslu.ch)
Zada Pajalic (zada.pajalic@hioa.no)
Anna Dencker (anna.dencker@gu.se)

Version: 1 Date: 03 Nov 2018

Author’s response to reviews:

We have done all required changes.

As this response comes really late in relation to submission date almost five months later we look forward to a very soon processing.

The order of documents is not optimal. I did not come as we want it in order to be logically following how they appear in the final manuscript of all merging documents. This as the system
order them: Manuscript, Tables and Figures. Also the system makes repetitions after References, thus these are documents that should not be included in the final version.